Author's response to reviews

Title: Lower respiratory tract infection and rapid expansion of an abdominal aortic aneurysm: A case report

Authors:

Steven Naylor (stevennaylor@doctors.org.uk)
Zakareya Gamie (ugm1zg@doctors.org.uk)
Ravinder S Vohra (raysvohra@hotmail.com)
Sapna Puppala (Sapna.Puppala@leedsth.nhs.uk)
Patrick J Kent (Patrick.Kent@leedsth.nhs.uk)
D Julian A Scott (profdjascott@yahoo.co.uk)

Version: 5 Date: 25 June 2010

Author's response to reviews:

Dear Editor,

We have made the below changes:

Major revisions

1/ All the references are now cited in the text using consecutive numbers in square brackets.

Minor revisions

1/ All underlining has been removed.

2/ The keywords section has been removed.

3/ The abbreviations, competing interests, and authors' contributions sections are now immediately before the reference list.

4/ Typography: We have corrected the Typography where possible.

Kind regards,

Dr Zakareya Gamie